Accelerator Status Report

Run Schedule: Jan. 3, 2011 through May 13, 2011
Feb. 23, 2011
Injector    Energy 61.76 MeV
· (2/22) The polarization flipper was changed from ‘flip’ left to ‘flip’ right. The injector was optimized.
· (2/21) Control modules of cavities 0L04-5 & 8 were replaced. The 45MeV spectrometer was used to phase the cavities and the spectrometer was left on (~6 hrs.).

· (2/17) Both Capture 2.5W Amplifiers were replaced. After the replacement, no adjustment was needed to transport beam (~1.5 hrs.). 
· (2/15) Beam studies: matching the injector to improve Hall C beam quality. After the match all three beam were different. The injector did not reproduce. 
· (2/14) Capture water skid tripped off once. We have the procedure to reset it.

· (2/14) The laser spot was moved to boost the QE.

· (2/11) The injector was optimized to improve the beam quality (mainly for Hall C halo and Compton counts).

· (2/10) The vertical polarization was backed out and a spin dance was performed. The HWien was set to -62.4 degrees. 

· (2/10) The 45MeV spectrometer power line was re-connected. The magnet is still off.
Accelerator   Linac Energy 549.00 MeV
· The main machine has been stable.

· (2/22) ITV6E02 failed and need to be repaired.
· (2/22) CHL tripped around 1100 due to a user error during CHL maintenance and tripped again around 1540 (about a shift was lost).

· (2/20) Arc2A current was not stable. The regulator board was replaced (1.67 hrs.).
· (2/20) MAB6T01 mismatched. It was reset OK. It affected pass 3 beam (Hall A beam). The mismatch was not identified until 7 hours later (0.8 hrs.). Do not silent an alarm unless you are sure what it is and what action you should take.
· (2/17) Lambertson viewer did not show the video and the camera was replaced. (1.4 hrs.)

· (2/17) VBV1L13A would not open and the position indicator was replaced (1.6 hrs.)

· (2/16) iocea3 controller and the FSD card for 3E01 node were replaced (2 hrs.)
· Main machine has been stable.
· (2/15) Pass 1 separator power dropped to zero and EES repaired it.

· (2/12) 2L05 waveguide pressure line leak and was fixed.  Cavity 2L05-2 control module was replaced.
· (2/12) Cavity 0L03-2 became unstable: the gradient was lowered to 3MeV. There were a lot if 2T09 BLM trips. Lowering the gradient of cavity 0L03-2 and steering 2T and 4T reduced the BLM trips.

· (2/10) Pass 1 separator was repaired and is in use. Pass 2 separator is for pass 2 now.

· (2/10) The DAC for 1L02 and 1L03 heater control was replaced.

· When to send beam after a vacuum event: whenever the beam line valves can be opened we can send beam.

Hall A---Pass 3, E08 - 010
· Beam is ggod. The hall will soon need high current (> 100uA).

· (2/17) Changed to pass 3. We forgot to initialize the fast raster and calibrate the ion chambers (they were taken care of three shifts later).
· Beam does not go through the Compton for this run.
· New Hall A Moller procedure.

· Will go to pass 3 Thursday.
· (2/12) Started to take low current beam (< 5uA).
Hall B ---Pass 5, TPE
· Beam is good. The hall will finish its current run Friday and HDice will start on March 26.

· (2/18) Changed to pass 5.

· (2/17) Changed to pass 2.
· Will go to pass 2 Thursday for a couple of shifts.

· The beam is good.
Hall C--- Pass 1, QWeak 
· Beam is good.

· (2/22)The width of Hall C’s beam charge asymmetry was larger than the specification. The optimization of the injector improved the beam quality.

· The hall started after the vacuum fix but the vacuum is not stable. They may ask beam to be off periodically to recover the vacuum.

· (2/18) Hall C was down for more than 5 shifts due to its vacuum issues at the end of its beam line (right upstream of its beam dump). 
· The beam may not be as good as before the injector match.
· The hall has vacuum issues in the dump area. It is not beam related.

